Author’s response to reviews

Title: Correlation between the Charlson comorbidity index and skeletal muscle mass/physical performance in hospitalized older people potentially suffering from sarcopenia

Authors:

Ge Gong (gonggedoc@163.com)
Wenhui Wan (356133926@qq.com)
Xinghu Zhang (zhangxh197002@163.com)
Yu Liu (liuyufanwei@163.com)
Xinhui Liu (linxinhuijnyy@163.com)
Jian Yin (yinjiandoc@163.com)

Version: 1  Date: 21 Aug 2019

Author’s response to reviews:

We have modified it as required and marked it in red in the article. Thank you.